Author's response to reviews

Title: Developing an Electronic Health Record (EHR) for Methadone Treatment Recording and Decision Support

Authors:

Liang Xiao (liangxiao@rcsi.ie)
Gráinne Cousins (gcousins@rcsi.ie)
Brenda Courtney (brendacourtney@rcsi.ie)
Lucy Hederman (hederman@scss.tcd.ie)
Tom Fahey (tomfahey@rcsi.ie)
Borislav D Dimitrov (borislavdimitrov@rcsi.ie)

Version: 6 Date: 27 January 2011

Author's response to reviews:

According to your instructions, we made all requested changes and we revised the manuscript and figures, accordingly. In particular, a separate page with the legends to figures was provided, reference list has been shortened, reviewed and formatted.